Author’s response to reviews

Title: Impact of stakeholder perspectives on cost-effectiveness estimates of four specialized nutritious foods for preventing stunting and wasting in children 6-23 months in Burkina Faso

Authors:

Ye Shen (ye.shen@tufts.edu;yeshen92@gmail.com)
Ilana Cliffer (Ilana.Cliffer@tufts.edu)
Devika Suri (Devika.Suri@tufts.edu)
Breanne Langlois (Breanne.Langlois@tufts.edu)
Stephen Vosti (vosti@primal.ucdavis.edu)
Patrick Webb (Patrick.Webb@tufts.edu)
Beatrice Rogers (beatrice.rogers@tufts.edu)

Version: 2 Date: 11 Feb 2020

Author’s response to reviews:

Dear Associate Editor,

Thanks so much for the three remaining comments. We have addressed them in the revision:

-- We have changed to "6 ~23" in line 125 and throughout.

-- Line 149 We have revised this sentence accordingly.

-- The two unpublished papers cited in this manuscript are recently published. Therefore, we have changed to the official citations in the references.

We would like to thank the editor and reviews again for this review process. Please let us know if you have any other questions or concerns.

Sincerely

Ye Shen